Author’s response to reviews

Title: Unusual Causes of Parkinsonism: a case series of potential cause-effect relationship

Authors:

Solomon O Ugoya (docsouls@yahoo.com)
Emmanuel I Agaba (eiagaba@yahoo.com)
Comfort A Daniyam (comfort_13@yahoo.com)

Version: 5 Date: 3 November 2010

Author’s response to reviews: see over
The Editor,

JMCR

Attention: Editorial Team

RE: MS: 6129007063691597- Submission of Manuscript titled: Unusual causes of parkinsonism: a case series

We are submitting the corrected version (galley proof) of our paper for possible publication in your reputable journal.

The suggested corrections are as follows: to change title to; Unusual causes of parkinsonism: a case series of potential cause-effect relationship.

- All the necessary typographical errors are made and changes made
- The references in the text has been made to journal style.
- The ethnicity of both cases has been added to the abstract and text.
- Tracked changes is accepted and the balloons removed
- Thank you for your effort in reading our manuscript.
- All authors had read and approved the recent changes.

Yours sincerely,

Dr. Solomon O. Ugoya